Author's response to reviews

Title: Dynamic 3D computed tomography for non-invasive identification of femoroacetabular impingement: cadaver validation study

Authors:

Maarten A Röling (maartenroling@hotmail.com)
Monique I Visser (moniqueingeborg@hotmail.com)
Edwin Oei (e.oei@erasmusmc.nl)
Peter Pilot (pilot@rdgg.nl)
Gert-Jan Kleinrensink (g.kleinrensink@erasmusmc.nl)
Rolf M Bloem (bloem@rdgg.nl)

Version: 2 Date: 4 March 2014

Author's response to reviews: see over
Cover letter revision manuscript: “Dynamic 3D computed tomography for non-invasive identification of femoroacetabular impingement: cadaver validation study”

Dear editors,

On behalf of my co-authors, I send you the revision of our manuscript. We have taken your comment seriously and hope that the changes made fulfill your demands. We listed each individual point that was marked by the editors.

1) Please reformat the title page at the front of your manuscript file. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
   - We apologize for the title page not to be in the right format. We added the missing information of the page.

2) Please note that Research article require the following sections: Background, Methods, Results, Discussion (Results and Discussion may be combined), Conclusions, List of abbreviations used (if any).
   - The sections were adjusted and the list of abbreviations was added to the title page. The abbreviations were already listed in the manuscript.

3) Research involving human subjects, human material, or human data must have been performed in accordance with the Declaration of Helsinki (http://www.wma.net/en/30publications/10policies/b3/index.html) and must have been approved by an appropriate ethics committee.
   - For this cadaveric validation study, no approving of a ethics committee was obliged in the Netherlands. All specimens were bodies donated to science, and no further committee had to approve this study.

4) By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship.
   - The “Acknowledgments” were added to the manuscript on page 8. Also, a statement about funding was added.

All additions and corrections were highlighted in the document and saved as “manuscript validation revision”.

We thank you for your comments and hope to hear from you.

Sincerely,

Maarten A. Röling.
Corresponding author
Maarten A. Röling, MD
Postal address:
Department of Orthopaedic surgery
Reinier de Graafweg 3-11
2526 AD Delft
the Netherlands